Author's response to reviews

Title: Age at Menarche in relation to nutritional status and critical life events among girls in Urban and Rural Secondary Schools in Post Conflict Northern Uganda

Authors:

Beatrice Odongkara Mpora (beachristo2003@yahoo.com)
Tereza W Piloya (tpiloa@yahoo.com)
Edison A Mworozi (arwanire@yahoo.com)
Paul Laigong (drlaigongp@yahoo.com)
Thomas Ngwiri (tngwiri@gerties.org)
Hochberg Ze'ev (z_hochberg@rambam.health.gov.il)

Version: 2 Date: 2 April 2013

Author's response to reviews:

Dear Editorial Team,

The revised manuscript contains all the components of the the first two sent earlier with addition of critical life events.

The authors are seven instead of six. I was unable to access the author page to incorporate 7th authors.

Thanks for giving us the opportunity to publish in your journal.

Dr Odongkara Beatrice Mpora
Corresping Author & PI